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W22 progress and plan
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Next step: 

CM04?
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CM01 arrival
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On going

the ferry between Germany and Sweden was delayed

yesterday (it left only at 23:30 instead of 16:30) and that the 

truck will be delivered at 14:00. No info to UU directly

wrong
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CM03: cooling down rate
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LHe level

GHe vapor flow

temperature sensors

CV03 Vbox’s filling valve

150 K (12:06h) < TT06 and TT07 < 50 K (12:55)

 rate=2.04 K/min 

26 hours for stable operation at 4K



CM03: coupler cold conditioning 
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beam vacuum

No problems



CM03: CTS thermalization over weekend
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Start N2 cooling May 26 evening

Reached around 100K May 31 morning
4 days



CM03: 2K pumping
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CV551 pumping valve

GHe vapor flow

LHe level

He pressure



CM03: frequency vs pressure
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CAV IN

14.3 Hz/mbar 15.5 Hz/mbar

CAV OUT

No problems



CM03: cavities without CTS engaged
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CAV IN

CAV OUT

f0 [MHz] QL

CAV IN 352.133 1.9e5

CAV OUT 352.116 1.9e5
OK for both



CM03: CAV IN performance
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CM03: CAV OUT performance
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No field emission



Static heat load measurements
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Fomr GHe vapor flow

June 2nd 9:30 14.31 W

June 3rd 9:30 14.69 W

We are also trying pressure rise method

Dynamic heat load measurement is ongoing right now 

but is promisingly good i.e. maybe not visible

• Using the DC heaters to calibrate the 

pressure rise rate

• Positive correlation is observed but 

needs to be investigated more

GHe flow is 0 (closed)
Cf. CM05 static: 

0.02376 mbar/s



CM03: CTS2 OK
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target 

352.21 MHz



CM03: CTS2 piezos OK
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PZ20: 470Hz

PZ21: 430Hz

unipolar 0-200V

bipolar -40 - +200V

PZ20: 620Hz

PZ21: 550Hz

unipolar 0-200V

bipolar -40 - +200V

PZ20+PZ21: 660Hz

PZ20+PZ21: 940Hz



CM03: CTS1 piezos maybe OK
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PZ10: 550Hz

PZ11: 480Hz

unipolar 0-200V

bipolar -40 - +200V

PZ20: 630Hz

PZ21: 550Hz

unipolar 0-200V

bipolar -40 - +200V

PZ10+PZ11: 910Hz

PZ10+PZ11: 

1070Hz



CM03: CTS1 behavior

• With 0.6A, first limit switch found at about -30. (led off) move in opposite 

direction several times led back on at maybe -10. Go negative again and 

can now not find the switch for -200.

• With 1.2A, it moved again but stuck as shown in the last slide

• Different from the issues in the first CM02 test, because it did not move 

even before the first engagement to the cavity

• There looks like a friction from the beginning
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Test plan: 3 working days reserved
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take frequency during 

disengagement

Use 1.2A from 

the beginning

decision making: 

return to Orsay?


